Author’s response to reviews

Title: The effects of oral administration of Cola nitida on the pharmacokinetic profile of metoclopramide in rabbits

Authors:

Cecilia Amadi (cnamadi@rocketmail.com)
Wisdom Nwachukwu (donspotless20.nw@gmail.com)

Version: 5 Date: 15 Dec 2019

Author’s response to reviews:

I have revised the indicated sections in the background, method and discussion sections.


I have also removed the coloured texts, response to reviewers’ comments, and ARRIVE guidelines checklist.